Accelerator Status Report
Run Schedule Jan. 14, 2009 to March 12, 2009

· Injector energy 65.64 MeV
· Linac Setup 583. 5 MeV

· Pass setup: 5/5/4
· The accelerator has been stable.

· The trim card for MFA0I03 was replaced, which made Hall A Compton rates more stable.

· The accelerator has been running well. 

· The optics has been stable. RF has been stable

· Arc4 box supply mismatched a few times causing several hours of downtime. EES fixed the problem.

Hall A
E06-010 5.9GeV/5 Pass
Measurement of Single Target-Spin Asymmetry
· Moller quad MQO1H03A has problems when restored from a Moller run.
· Chimes have been running to make sure Hall A beam position is within -/+2mm window of its nominal position.

· Ops still need pay attention to the Compton rates.

· There were some problems restoring beam after Hall A Moller runs Saturday.

· The Moller procedures has been updated.

Hall B
· The hall is down. They finished their experiment.
· The hall is planning to come back on Thursday swing shift (2/5).
· The COO is available in the MCC.
Hall C

E07-003-SANE

4.734/4 Pass 

· Hall C had a lot of downtime due to its target related problems.

· The target coil FSD read back is not stable. We can adjust the FSD window to clear the fault.
· The hall has been back online since Saturday. The setup went well. 
· The hall still has problems with its target.

· Current <=0.2 uA

· Use the PID lock to control the current.

· FFB Status: FFB off.

